Author's response to reviews

Title: Human Rights, health and the state in Bangladesh School of Public Health, La Trobe University, Melbourne, Victoria-3086, Australia. and Department of Political Studies & Public Administration, Shah Jalal University of Science & Technology, Sylhet, 3114, Bangladesh. Email: rahmanredwan@yahoo.com

Authors: Redwanur MR Rahman (rahmanredwan@yahoo.com)

Version: 7 Date: 13 December 2005

Author's response to reviews:

Affiliations are placed, keywords are removed and figure is deleted. Figure and table placement text are also deleted